Author’s response to reviews

Title: An atypical form of AOA2 with myoclonus associated with mutations in SETX and AFG3L2

Authors:

Cecilia Mancini (cecilia.mancini@unito.it)
Laura Orsi (lauraors11@hotmail.com)
Yiran Guo (GUOY@email.chop.edu)
Xuanzhu Liu (liuxuanzhu@genomics.cn)
Yi Xie (xieyi@genomics.cn)
Fengxiang Wang (wangf@email.chop.edu)
Lifeng Tian (tianl@email.chop.edu)
Wenfei Chen (chenwenfei@genomics.cn)
Jinlong Liang (liangjinlong@genomics.cn)
Bruce Nmezi (bcn6@pitt.edu)
Takahashi Tatsuta (takashi.tatsuta@uni-koeln.de)
Elisa Giorgio (elisa.giorgio84@libero.it)
Eleonora Di Gregorio (eleonora.digregorio@unito.it)
simona Cavalieri (scavalieri@gmail.com)
elisa pozzi (elisa.pozzi@unito.it)
paolo mortara (paolo.mortara@unito.ito.it)
maira m caglio (marcella.caglio@unito.it)
alessandro balducci (alessandro.balducci@unito.it)
lorenzo pinessi (lorenzo.pinessi@unito.it)
thomas langer (thomas.langer@uni-koeln.de)
quasar s padiath (jianghui@genomics.cn)
hakon hakonarson (hakonarson@chop.edu)
hui lang (gpadiath@pitt.edu)
alfredo brusco (alfredo.brusco@unito.it)

Version: 4
Date: 13 February 2015

Author’s response to reviews: see over
Dear Editor

Thank you for accepting our submission “An atypical form of AOA2 with myoclonus associated with mutations in SETX and AFG3L2” (MS: 1646497839150528) for publication in BMC Medical Genetics.
As requested, I obtained a specific signed consent for publication by all participants in the study.
Forms are in Italian, but based on an English version provided by the American Journal of Medical Genetics. A sentence has been added in materials and methods (highlighted).
I took the occasion to edit some typos and correct some sentences which were not completely clear (highlighted). Reading the text I realized that a page of materials and methods (genetic analysis) was in supplement, while it makes more sense to be in the main manuscript.
Furthermore, some of the authors name were wrong in the previous version, because of an incorrect list from collaborators.
I hope I have answered all requests,

Yours Sincerely,

Alfredo Brusco
alfredo.brusco@unito.it